Author's response to reviews

Title: Genetic Polymorphisms of Angiotensin-2 Type 1 Receptor and Angiotensinogen and Risk of Renal Dysfunction and Coronary Heart Disease in Type 2 Diabetes Mellitus

Authors:

Julie Lin (jlin11@partners.org)
Frank B Hu (nhbfh@channing.harvard.edu)
Lu Qi (nhlqi@channing.harvard.edu)
Gary C Curhan (gcurhan@partners.org)

Version: 2 Date: 10 February 2009

Author's response to reviews: see over